Author's response to reviews

Title: Sunitinib Combined with Proton Beam Radiation for the Treatment of Chondrosarcoma: a Case Report

Authors:

Jennifer Dallas (jennifer.dallas@medicine.ufl.edu)
Iman Imanirad (iman.imanirad@medicine.ufl.edu)
Rajiv Rajani (rajanr@ortho.ufl.edu)
Roi Dagan (rdagan@ufl.edu)
Sukanthini Subbiah (Sukanthini.Subbiah@medicine.ufl.edu)
Rebecca Gaa (gaar@shands.ufl.edu)
Wayne A. Dwarica (Dwariwa@shands.ufl.edu)
Alison M. Ivey (aivey@ufl.edu)
Robert A. Zlotecki (zlotera@ufl.edu)
Robert Malyapa (malyapa@ufl.edu)
Danny J. Indelicato (dindelicato@floridaproton.org)
Mark T. Scarborough (scarbmt@ortho.ufl.edu)
John D. Reith (reith@pathology.ufl.edu)
C. Parker Gibbs (cpgibbs@ortho.ufl.edu)
Long H. Dang (long.dang@medicine.ufl.edu)

Version: 2 Date: 12 November 2011

Author's response to reviews: see over
November 6, 2011

Editorial Board of *Journal of Medical Case Reports*,

Dear Members of the Editorial Board,

We are submitting our manuscript titled “Sunitinib combined with proton beam radiation for the treatment of chondrosarcoma: a case report” for your consideration as a case report in *Journal of Medical Case Reports*.

We present a case of a patient with clear cell chondrosarcoma who presented with symptomatic recurrence in the pelvis and metastases to the skull and lungs, and was treated with sunitinib and proton beam radiation. Although sunitinib has shown efficacy in patients with various tumor types and is approved by the FDA for treatment of renal cell cancer (RCC) and gastrointestinal stromal tumors (GIST), there has been no reported experience in chondrosarcoma.

Thank you very much for your time and consideration.

Sincerely yours,

Long H. Dang MD PhD